

### New South Wales

# Sporting Venues Authorities Amendment (Venues NSW) Bill 2020

# **Explanatory note**

This explanatory note relates to this Bill as introduced into Parliament.

### Overview of Bill

The objects of this Bill are—

- (a) to amend the Sporting Venues Authorities Act 2008—
  - (i) to reconstitute the regional sporting venues authority known as Venues NSW and dissolve the Sydney Cricket Ground Trust, and
  - (ii) to provide for the establishment of advisory committees to provide advice to the Minister or Venues NSW and enable Venues NSW to exercise its functions, and
  - (iii) to provide for the ownership or management of land previously administered by the Sydney Cricket Ground Trust to be transferred to Venues NSW, and
  - (iv) to make provisions of a savings or transitional nature to give effect to the above matters,
- (b) to amend the *Sydney Olympic Park Authority Act 2001* to enable the Governor, by written order, to transfer assets, rights and liabilities from the Sydney Olympic Park Authority to another public sector agency,
- (c) to make other consequential or related amendments.

# Outline of provisions

Clause 1 sets out the name (also called the short title) of the proposed Act.

Clause 2 provides for the commencement of the proposed Act on a day or days to be appointed by proclamation.

Clause 3 repeals the Sydney Cricket and Sports Ground Act 1978 and the Sydney Cricket Ground and Sydney Football Stadium By-law 2014.

# Schedule 1 Amendment of Sporting Venues Authorities Act 2008 No 65

Schedule 1[1] inserts definitions for terms used in the amendments made to the Act. Schedule 1[8] and [9] make consequential amendments.

Schedule 1[4] specifies the objects of the Act, namely—

- (a) to enable the efficient and effective management and coordination of entertainment and sporting venues within the State on a commercial basis, and
- (b) to attract and encourage the holding of major events at entertainment and sporting venues within the State, and
- (c) to facilitate the development of precincts surrounding the entertainment and sporting venues that Venues NSW owns or operates, and
- (d) to ensure a customer-focused approach is applied to the management of entertainment and sporting venues and surrounding lands so they remain accessible to, and meet the needs of, users of and visitors to the venues and the broader community, and
- (e) to ensure the provision of entertainment and sporting venues fulfils community service obligations that may be determined by the Minister from time to time.

Schedule 1[5] removes a redundant delegation power of the State Sporting Venues Authority.

**Schedule 1[10]** repeals Part 3 of the Act and inserts new Parts 3 and 3A, which provide for the establishment of Venues NSW and advisory committees.

Under new Part 3—

- (a) proposed section 12 provides for the constitution of Venues NSW as a corporation, and
- (b) proposed section 13 states Venues NSW is a NSW Government agency, and
- (c) proposed section 14 provides Venues NSW is subject to the control and direction of the Minister, and
- (d) proposed section 15 establishes the Board of Venues NSW, and
- (e) proposed section 16 provides the Minister may appoint a Chief Executive Officer of Venues NSW in consultation with the Board, and
- (f) proposed section 17 provides Venues NSW or a private subsidiary corporation of Venues NSW may also employ its own staff, and
- (g) proposed section 18 confirms the doctrine of incompatibility of office does not operate to prevent a holder of a position in a government sector agency from being appointed to a position in Venues NSW or on its Board, and
- (h) proposed section 19 sets out the principal functions of Venues NSW, which include establishing and managing sporting grounds, sporting facilities and recreational facilities, providing advice to the Minister on the State's entertainment and sporting venues strategy and promoting the use of Venues NSW's land for commercial purposes, and
- (i) proposed section 20 confirms the ability of Venues NSW to delegate its functions to certain authorised persons, excluding the power of delegation, and
- (j) proposed section 21 enables Venues NSW to form or acquire interests in certain private corporations, and
- (k) proposed section 21A enables Venues NSW to exercise its functions through private subsidiary corporations, or in a partnership, joint venture or other association with other persons or bodies.

Under new Part 3A—

- (a) proposed section 21C enables Venues NSW to establish advisory committees to provide advice to the Minister or Venues NSW, or to enable Venues NSW to exercise its functions and proposed section 21B includes definitions used in the new Part, and
- (b) proposed section 21D—
  - (i) establishes membership advisory committees to provide advice to Venues NSW regarding the membership of Members of the Sydney Cricket Ground and persons who held membership to stadia formerly vested in the regional sporting venues authority known as Venues NSW, and
  - (ii) provides the membership of the membership advisory committees is to be made up of former members of the Sydney Cricket and Sports Ground Trust or the board of management of former Venues NSW, and
  - (iii) for the membership advisory committee established in respect of membership of the Sydney Cricket Ground, further members may be elected to the committee in accordance with a Schedule 3 to be inserted by **Schedule 1**[25].

Schedule 1[2], [3], [6], [7], [12] and [16] make consequential amendments.

**Schedule 1[11]** provides the transfer of land to a sporting venues authority under Part 4, Division 1 of the Act is conduct authorised for the purposes of the *Competition and Consumer Act 2010* of the Commonwealth and the *Competition Code of New South Wales*.

**Schedule 1[13]** confirms a sporting venues authority's ability to manage, develop and otherwise deal with its land is subject to the provisions of proposed Part 4, Divisions 2A and 2B inserted by Schedule 1[14].

**Schedule 1[14]** inserts proposed Part 4, Divisions 2A and 2B, which contain Crown land management and development provisions relating to Venues NSW's land previously vested in the Sydney Cricket and Sports Ground Trust.

Proposed Division 2A provides for the appointment of Venues NSW as the Crown land manager for land previously vested in the Sydney Cricket and Sports Ground Trust, described in the Act as *scheduled lands*. The Division also confirms the scheduled lands continue to be Crown land dedicated for the purposes of public recreation.

Proposed Division 2B makes particular provision in relation to scheduled lands, designated land and controlled land that are further described in Schedule 4A of the Act. Under proposed Division 2B—

- (a) proposed section 30AD—
  - (i) enables the Minister to approve the carrying out of development on designated land, along with plans and specifications associated with the development, and
  - (ii) requires the Minister, before giving the approval, to consult with the Minister or Ministers administering the *Public Works and Procurement Act 1912* and the *Environmental Planning and Assessment Act 1979*, and
- (b) proposed section 30AE provides certain environmental planning laws do not apply to the development approved by the Minister under proposed section 30AD, and
- (c) proposed section 30AF enables the scheduled lands to be the subject of a State environmental planning policy under the *Environmental Planning and Assessment Act* 1979 and further enables the land to be used for purposes permitted by the planning policy (subject to the restrictions in proposed section 30AI), and
- (d) proposed section 30AG enables the Minister to make a determination that an agreement, lease or licence relating to the scheduled lands may be varied, or cease to apply to a party, to enable work to be carried out on the scheduled lands, and
- (e) proposed section 30AH makes it clear the scheduled lands may be used for purposes permitted under a State environmental planning policy, and a lease or licence granted for

- the use of the land for those purposes, even though the land is dedicated for public recreation, and
- (f) proposed section 30AI prevents the use of any part of the scheduled lands other than controlled land for residential accommodation, and any part of the scheduled lands other than designated land for tourist and visitor accommodation, and
- (g) proposed section 30AJ provides the Minister may declare certain land within Venues NSW's land to be controlled land or designated land.

**Schedule 1[15]** omits Part 5 of the Act relating to the requirement for regional sporting venues authorities to develop plans of management. This Part is no longer necessary as there will no longer be any regional sporting venues authorities under the Act.

**Schedule 1[17] and [18]** provide Venues NSW is the authority required to allow the Newcastle Agricultural, Horticultural and Industrial Association to use the Showground land.

**Schedule 1[19] and [21]** provide the Chief Executive Officer of Venues NSW may appoint rangers for the purposes of the Act and delegate the Chief Executive Officer's appointment functions to a person employed by Venues NSW.

**Schedule 1[20]** makes it clear a ranger has the powers and duties conferred on a ranger by the Act or the regulations.

**Schedule 1[22]–[24]** enables the regulations to provide for additional matters relating to the functions of the State Sporting Venues Authority and Venues NSW.

Schedule 1[25] inserts proposed Schedules 1, 2 and 3 in the Act.

Proposed Schedule 1 provides for the constitution and procedures of the Board of Venues NSW.

Proposed Schedule 2 contains provisions regarding the appointment, remuneration and removal of the Chief Executive Officer.

Proposed Schedule 3 sets out the procedures for the election of members of the membership advisory committee regarding membership of the Sydney Cricket Ground.

**Schedule 1[26]** inserts Schedule 4A in the Act, which sets out the description of certain land vested in, or managed by, Venues NSW for the purposes of the definitions provided under section 3(1) of the Act.

**Schedule 1[27]** inserts Schedule 5, Part 6 in the Act, which contains savings, transitional and other provisions consequent on the enactment of the proposed Act. The Part includes provisions that—

- (a) provide for the reconstitution of Venues NSW, and
- (b) dissolve the Sydney Cricket Ground Trust and also Local Venues Councils constituted under the Act, and
- (c) provide for the transfer of the assets, rights and liabilities of the Sydney Cricket Ground Trust and Local Venues Councils, and certain regulatory authorisations held by those former bodies, to Venues NSW, and
- (d) provide for the transfer of employees of the Sydney Cricket Ground Trust and current employees of the Public Service to Venues NSW, and
- (e) confirm persons who were previously authorised for certain functions relating to admission to, and removal of persons from, lands under the *Sydney Cricket Ground and Sydney Football Stadium By-law 2014* before its repeal are taken to have been appointed as rangers under the Act.

# Schedule 2 Amendment of Sydney Olympic Park Authority Act 2001 No 57

**Schedule 2** enables the Governor, by written order, to transfer assets, rights and liabilities of the Sydney Olympic Park Authority to another public sector agency.

# Schedule 3 Amendment of other legislation

#### **Sporting Venues Authorities Regulation 2019**

**Schedule 3.7[1]** provides a ranger or police officer may direct a person to leave land or a facility vested in or managed by a sporting venues authority if the ranger or police officer considers the person is contravening a provision of the Act or the Regulation.

**Schedule 3.7[2]** confirms the capacity for Venues NSW or a ranger to call in the aid of a police officer to assist with the removal of a person from the land of a sporting venues authority.

Schedule 3.7[3] and [6] omit redundant provisions.

Schedule 3.7[4] updates references as a result of the amendments made by Schedule 1.

**Schedule 3.7[5]** provides certain term of office provisions do not apply to members of a membership advisory committee who were members of the Sydney Cricket and Sports Ground Trust or a member of the former board of management of Venues NSW immediately before the dissolution of those bodies.

#### Sporting Venues (Invasions) Regulation 2016

**Schedule 3.8** prescribes the Sydney Cricket Ground and Sydney Football Stadium as sporting venues to which the provisions of the *Sporting Venues (Invasions) Act 2003* apply.

#### Amendment of other legislation

**Schedules 3.1–3.6 and 3.9–3.14** make consequential amendments arising from the dissolution of the Sydney Cricket and Sports Ground Trust.



# New South Wales

# **Sporting Venues Authorities Amendment** (Venues NSW) Bill 2020

# **Contents**

			Page
	1	Name of Act	2
	2	Commencement	2
	3	Repeals	2
Schedule 1		Amendment of Sporting Venues Authorities Act 2008 No 65	3
Schedule 2		Amendment of Sydney Olympic Park Authority Act 2001 No 57	37
Schedule 3		Amendment of other legislation	40



## New South Wales

# **Sporting Venues Authorities Amendment** (Venues NSW) Bill 2020

No , 2020

#### A Bill for

An Act to amend the *Sporting Venues Authorities Act 2008* to reconstitute Venues NSW and dissolve the Sydney Cricket Ground Trust; to repeal and make consequential or related amendments to certain legislation.

The	The Legislature of New South Wales enacts—	
1	Name of Act	2
	This Act is the Sporting Venues Authorities Amendment (Venues NSW) Act 2020.	3
2	Commencement	4
	This Act commences on a day or days to be appointed by proclamation.	5
3	Repeals	6
	The following are repealed—	7
	(a) the Sydney Cricket and Sports Ground Act 1978 No 72,	8
	(b) the Sydney Cricket Ground and Sydney Football Stadium By-law 2014.	9

Scł	nedule 1	Amendment of Sporting Ver 2008 No 65	nues Authorities Act
[1]	Section 3 [	efinitions	
	Insert in alp	abetical order in section 3(1)—	
		advisory committee means a committee esta includes a membership advisory committee.	ablished under section 21C and
		<b>Board of Venues NSW</b> means the Board of V 15.	enues NSW provided by section
		Chief Executive Officer means the Chief Exe	ecutive Officer of Venues NSW.
		controlled land means the lands described in	Schedule 4A, Part 3.
		designated land means the land described in	Schedule 4A, Part 2.
		<i>membership advisory committee</i> means a mestablished by section 21D.	nembership advisory committee
		<i>private corporation</i> means a corporation <i>Corporations Act 2001</i> of the Commonwealth Wales.	
		private subsidiary corporation means a p sporting venues authority has a controlling in	
		scheduled lands means the land described in	Schedule 4A, Part 1.
		Venues NSW means Venues NSW constitute	d by this Act.
		<b>Venues NSW's land</b> means the land vested is and includes land described in Schedule 4A.	n or managed by Venues NSW
[2]	Section 3(1 authority"	, definitions of "board of management" an	d "regional sporting venues
	Omit the de	initions.	
[3]	Section 3(1	, definition of "sporting venues authority"	
	Omit "a reg	onal sporting venues authority". Insert instead	"Venues NSW".
[4]	Section 3A		
	Insert after	ection 3—	
	3A Obje	ts of Act	
		The objects of this Act are as follows—	
		(a) to enable the efficient and effective m entertainment and sporting venues with basis,	
		(b) to attract and encourage the holding o and sporting venues within the State,	f major events at entertainment
		(c) to facilitate the development of precinct and sporting venues that Venues NSW	
		(d) to ensure a customer-focused approach entertainment and sporting venues a remain accessible to, and meet the need venues and the broader community,	nd surrounding lands so they

		(e)	to ensure the provision of entertainment and sporting venues fulfils community service obligations that may be determined by the Minister from time to time.	1 2 3
[5]	Sect	ion 7 Dele	gation of State Sporting Venues Authority's functions	4
	Omi	section 7(	3)(b) and (c). Insert instead—	5
		(b)	a government agency or member of staff of a government agency.	6
[6]	Sect	ion 9 Func	tions of State Sporting Venues Authority	7
	Omi	"a regiona	l sporting venues authority" from section 9(2).	8
	Inser	t instead "V	Venues NSW".	9
[7]	Sect	ion 9(2)		10
	Omi	t "the region	nal sporting venues authority". Insert instead "Venues NSW".	11
[8]	Sect	ion 10 Exe	rcise of functions through private subsidiaries, joint ventures etc	12
	Omi	t "(within tl	ne meaning of section 11)" from section 10(b).	13
[9]	Sect	ion 11 Priv	rate subsidiary corporations etc	14
	Omi	section 11	(1).	15
[10]	Part	3		16
	Omi	the Part. In	nsert instead—	17
	Par	t 3 Ve	enues NSW	18
	Divi	sion 1	Constitution and administration of Venues NSW	19
	12	Constitu	tion of Venues NSW	20
		The Ve	ere is constituted by this Act a corporation with the corporate name of nues NSW.	21 22
	13	Status of	Venues NSW	23
		Ve	nues NSW is a NSW Government agency.	24
	14	Ministeri	al control	25
		exe	nues NSW is subject to the control and direction of the Minister in the ercise of its functions, except in relation to the contents of any advice, report recommendation given to the Minister.	26 27 28
	15	Board of	Venues NSW	29
		(1) The	ere is to be a Board of Venues NSW.	30
			e Board of Venues NSW is to consist of the following members appointed the Minister—	31 32
		(a)		33
		(b)	Board,	34 35
		(c)	1 person who is a Public Service senior executive within the meaning of the Government Sector Employment Act 2013.	36 37

		(d) Note	at least 6, but not more than 8, other persons.  Schedule 1 sets out provisions relating to the members and procedure of the	1 2
			d of Venues NSW.	3
16	Chie	f Exec	cutive Officer of Venues NSW	4
	(1)		Minister may, in consultation with the Board of Venues NSW, appoint a f Executive Officer of Venues NSW.	5 6
	(2)	the a	Chief Executive Officer is responsible for the day-to-day management of affairs of Venues NSW in accordance with the policies and general tions of the Board of Venues NSW.	7 8 9
	(3)		act, matter or thing done in the name of, or on behalf of, Venues NSW by Chief Executive Officer is taken to have been done by Venues NSW.	10 11
	(4)	meet but i	Chief Executive Officer may attend, and participate in discussions at, ings of the Board of Venues NSW unless otherwise directed by the Board, s not entitled to vote at the meetings.  Schedule 2 sets out further provisions relating to the Chief Executive Officer.	12 13 14 15
17	Staff	of Ve	nues NSW	16
	(1)	Venu	ues NSW or a private subsidiary corporation may employ its own staff.	17
	(2)	emp	provisions of the <i>Government Sector Employment Act 2013</i> relating to the loyment of Public Service employees do not apply to staff employed by use NSW or a private subsidiary corporation.	18 19 20
18	Hold	er of o	office not affected by appointment to Venues NSW or its Board	21
	(1)	The	doctrine of incompatibility of office—	22
		(a)	does not operate to prevent the holder of a position in a government sector agency (the <i>original position</i> ) from being appointed to a position in Venues NSW or on its Board (the <i>additional position</i> ), and	23 24 25
		(b)	does not operate to effect or require the holder of the original position to surrender or vacate the position as a result of the appointment to the additional position.	26 27 28
	(2)		section applies even if the original position or the additional position is on an acting or temporary basis.	29 30
	(3)	This Act 2	section does not limit section 73 of the Government Sector Employment 2013.	31 32
	(4)	gove	is section— <b>rnment sector agency</b> has the same meaning as in the <i>Government Sector loyment Act 2013</i> .	33 34 35
Divi	sion	2	Functions of Venues NSW	36
19	Fund	ctions	of Venues NSW	37
	(1)	The	principal functions of Venues NSW are as follows—	38
		(a)	to provide advice to the Minister on the State's entertainment and sporting venues management strategy and development of existing or new entertainment and sporting venues and their surrounding precincts,	39 40 41
		(b)	to maintain and improve Venues NSW's land,	42
		(c)	to establish, manage and improve sporting grounds, events grounds and associated facilities for the purposes of sports, events and recreation.	43 44

	(d)	to establish, manage and improve facilities for the purposes of sporting competitions, sports training and sports education,	1 2
	(e)	to establish, manage and improve community facilities and to establish, manage and improve facilities for community and recreational purposes,	3 4 5
	(f)	to permit the use of the whole or any part of Venues NSW's land for activities of a sporting, recreational or community nature, including the use of the land for events and general community access,	6 7 8
	(g)	to encourage, if Venues NSW considers it appropriate, the use and enjoyment of Venues NSW's land by the public and clubs, associations or other bodies,	9 10 11
	(h)	to collaborate with public sector agencies and other bodies to promote the use of Venues NSW's land and facilities for commercial purposes,	12 13
	(i)	to undertake or provide, or facilitate the undertaking or provision of, commercial and retail activities and facilities on Venues NSW's land,	14 15
	(j)	if the Minister considers it appropriate, to undertake or provide, or facilitate the undertaking or provision of, residential or tourist and visitor accommodation on Venues NSW's lands designated for the purpose,	16 17 18 19
	(k)	to make all reasonable attempts to ensure new development carried out on Venues NSW's land accords with environmental and planning laws, standards and best practice,	20 21 22
	(1)	to contribute to proposals for the design, development and management of land neighbouring Venues NSW's land,	23 24
	(m)	to meet the community service obligations determined from time to time by the Minister in relation to the provision of entertainment and sporting venues.	25 26 27
(2)	may enter	ues NSW may undertake supplemental, incidental or consequential acts as be necessary or expedient for the exercise of its functions, including ring into any contract or arrangement with any person in connection with arrying out of its functions.	28 29 30 31
(3)		ues NSW has other functions conferred or imposed on it by or under this or any other Act.	32 33
Dele	gation	of Venues NSW's functions	34
(1)		ues NSW may delegate to an authorised person any of its functions, other this power of delegation.	35 36
(2)	A de by V NSW	elegate may sub-delegate to an authorised person any function delegated venues NSW if the delegate is authorised in writing to do so by Venues V.	37 38 39
(3)	In th	is section, authorised person means—	40
	(a)	a member of staff of Venues NSW, or	41
	(b)	a government agency or member of staff of a government agency, or	42
	(c)	an advisory committee or member of an advisory committee.	43
Priva	ate su	bsidiary corporations	44
(1)	Venu	ues NSW may—	45
	(a)	form, or participate in the formation of, private corporations, and	46

		(b)	acquire interests in private corporations, and	1
		(c)	sell or otherwise dispose of interests in private corporations.	2
	(2)	A pr	ivate subsidiary corporation is not, and does not represent, the Crown.	3
21A			f functions through private subsidiary corporations or with other r bodies	4 5
		A fu	nction of Venues NSW may be exercised—	6
		(a)	by Venues NSW itself, or	7
		(b)	by a private subsidiary corporation, or	8
		(c)	by Venues NSW or a private subsidiary corporation, or both, in a partnership, joint venture or other association with other persons or bodies.	9 10 11
Par	t 3A	Adv	visory committees	12
21B	Defir	nitions	•	13
	Dom		is Part—	
			mencement day means the day on which this Part commences.	14 15
			e includes omitted to be done.	16
			<i>lity</i> means civil liability and includes action, claim and demand.	17
21C	Advi	sory (	Committees—general	18
	(1)	Venu	ues NSW may establish advisory committees to—	19
	. ,	(a)	provide advice to the Minister or to Venues NSW, or	20
		(b)	enable Venues NSW to exercise its functions.	21
	(2)		ect to any directions given by the Minister, the procedure of an advisory mittee is to be determined by the Board of Venues NSW.	22 23
	(3)	remu	nember of an advisory committee is an honorary member and no ineration is payable to the member in respect of the duties they perform as imber of the advisory committee.	24 25 26
	(4)		dvisory committee, other than a membership advisory committee, may be olved or merged with another advisory committee at any time.	27 28
	(5)	The	regulations may provide for the following—	29
		(a)	the establishment and functions of an advisory committee,	30
		(b)	the membership of an advisory committee,	31
		(c)	the procedure and dissolution of an advisory committee,	32
		(d)	the appointment, term of office and removal from office of members of an advisory committee.	33 34
21D	Mem	bersh	ip advisory committees	35
	(1)		following membership advisory committees are established on the mencement day—	36 37
		(a)	a committee to provide advice to Venues NSW regarding the membership of persons who were formerly Members of the Sydney Cricket Ground, as referred to in clause 37 of the Sydney Cricket Ground and Sydney Football Stadium By-law 2014 before its repeal,	38 39 40 41

	(b)	a committee to provide advice to Venues NSW regarding the membership of persons who were given membership to stadia vested in Venues NSW before the commencement day.	1 2 3
(2)	Venu of—	nes NSW is to ensure a former member is invited to become a member	4 5
	(a)	if the former member was a member of the Sydney Cricket and Sports Ground Trust—the membership advisory committee established under subsection (1)(a), or	6 7 8
	(b)	if the former member was a member of the former board of management of Venues NSW—the membership advisory committee established under subsection (1)(b).	9 10 11
(3)	A pe	rson becomes a member of a membership advisory committee—	12
	(a)	if the person is a former member—	13
		(i) on the day the former member accepts Venues NSW's invitation to become a member of the membership advisory committee, and	14 15
		(ii) in accordance with subsection (2), or	16
	(b)	if the person who is elected, in accordance with Schedule 3, as a member of the membership advisory committee established under subsection (1)(a)—on the date of the person's election.	17 18 19
(4)		embership advisory committee may be dissolved or merged with another sory committee on the expiration of the remainder of the terms of office	20 21 22
	(a)	for a membership advisory committee established under subsection (1)(a)—all of the former members of the Sydney Cricket and Sports Ground Trust who have been appointed as members of the membership advisory committee, or	23 24 25 26
	(b)	for a membership advisory committee established under subsection (1)(b)—all of the former members of the former board of management of Venues NSW who have been appointed as members of the membership advisory committee.	27 28 29 30
(5)		ember of a membership advisory committee is not personally subject to iability for anything done—	31 32
	(a)	in good faith, and	33
	(b)	for the purpose of executing functions under this Act.	34
(6)	The 1	liability instead attaches to the Crown.	35
(7)	In thi	is section—	36
	mana	er board of management of Venues NSW means the board of agement of Venues NSW before Venues NSW was reconstituted by dule 5, Part 6.	37 38 39
	form	er member means a person who—	40
	(a)	was, immediately before its dissolution, a member of—	41
		(i) the Sydney Cricket and Sports Ground Trust, or	42
		(ii) the former board of management of Venues NSW, and	43
	(b)	is not appointed as a member of the Board of Venues NSW.	44
Section 25	Α		45
Insert after	section	n 25—	46

[11]

	25A	Competition authorisations relating to transfer of land under this Division					
		(1)	spec Cons	duct authorised or required by or under the provisions of this Division is ifically authorised by this Act for the purposes of the <i>Competition and sumer Act 2010</i> of the Commonwealth and the <i>Competition Code of New h Wales</i> .	2 3 4 5		
		(2)	wou	duct authorised by this section is authorised only to the extent (if any) it ld otherwise contravene Part IV of the Competition and Consumer Act of the Commonwealth and the Competition Code of New South Wales.	6 7 8		
[12]	Part	4 Prov	ision	s relating to vesting of and dealings with land and other property	9		
	Omi	t "A reg	gional	sporting venues authority" wherever occurring in Division 2.	10		
	Inser	t instea	ıd "Ve	enues NSW".	11		
[13]	Sect	ion 30	Rese	rvations, dedications and original grants of land	12		
	Inser	t at the	end o	of the section—	13		
		(2)	This	section is subject to Divisions 2A and 2B.	14		
[14]	Part	4, Divi	sions	s 2A and 2B	15		
	Inser	Insert after Division 2—					
	Divi	sion	2 <b>A</b>	Crown land management	17		
;	30AA	A Definitions					
			In th	is Division—	19		
			Crov	wencement day means the day on which this Division commences.  we land Minister means the Minister administering the Crown Land agement Act 2016.	20 21 22		
;	30AB	Dedi	catior	n of scheduled lands	23		
			Crov	scheduled lands continue, on and from the commencement day, to be wn land dedicated for the purpose of public recreation under the <i>Crown d Management Act 2016</i> .	24 25 26		
;	30AC	Venu	es NS	SW is Crown land manager of scheduled lands	27		
		(1)	On a	and from the commencement day—	28		
			(a)	the appointment of the Sydney Cricket and Sports Ground Trust as Crown land manager for the scheduled lands under the <i>Crown Land Management Act 2016</i> is taken to be revoked, and	29 30 31		
			(b)	Venues NSW is taken to have been appointed under the <i>Crown Land Management Act 2016</i> as the Crown land manager for the scheduled lands.	32 33 34		
		(2)	land	Crown Land Management Act 2016 applies in relation to the scheduled s and to Venues NSW in its capacity as the Crown land manager, subject e following modifications—	35 36 37		
			(a)	Venues NSW's appointment as the Crown land manager of the scheduled lands cannot be revoked under the <i>Crown Land Management Act 2016</i> ,	38 39 40		
			(b)	no other person can be appointed under the <i>Crown Land Management Act 2016</i> as a Crown land manager of the scheduled lands,	41 42		

		(c)	Venues NSW is taken to have been assigned as a category 1 non-council manager of the scheduled lands for the purposes of Division 3.5 of the <i>Crown Land Management Act 2016</i> ,	1 2 3
		(d)	if the Minister is not also the Crown land Minister, the Minister can—	4
			(i) exercise the functions of the Crown land Minister under Part 2 of the <i>Crown Land Management Act 2016</i> instead of the Crown land Minister in relation to the scheduled lands, except a function prescribed under paragraph (e), and	5 6 7 8
			(ii) grant written consent for the purposes of Division 3.5 of the <i>Crown Land Management Act 2016</i> ,	9 10
		(e)	the regulations under this Act may prescribe kinds of functions for the purposes of section 3.26(2)(d) of the <i>Crown Land Management Act</i> 2016,	11 12 13
		(f)	Venues NSW is not required to obtain the Minister's consent for the exercise of a function prescribed for the purposes of section 3.26(2)(d) of the <i>Crown Land Management Act 2016</i> or paragraph (e),	14 15 16
		(g)	the following provisions of the <i>Crown Land Management Act 2016</i> do not apply in relation to the scheduled lands or Venues NSW—	17 18
			(i) Division 2.2,	19
			(ii) Division 2.4 (except for sections 2.12 and 2.14),	20
			(iii) section 2.21,	21
			(iv) Division 3.2,	22
			(v) sections 3.14–3.16 and 3.18,	23
			(vi) sections 3.29–3.31,	24
			(vii) Division 3.6,	25
		(	(viii) section 3.45,	26
			(ix) sections 9.12 and 9.25,	27
			(x) Schedule 5,	28
			(xi) any other provisions prescribed by the regulations.	29
	(3)	this s	e Minister exercises a function of the Crown land Minister permitted by section, the exercise of the function has the same effect as if it had been exercised by the Crown land Minister.	30 31 32
Divi	sion	2B	Particular provisions relating to controlled land, designated land and scheduled lands	33 34
AD	Carr	ying o	ut of development on designated land	35
	(1)	The I	Minister may approve the following proposals—	36
		(a)	the carrying out of development on designated land,	37
		(b)	plans or specifications relating to the development.	38
	(2)		Minister's approval must certify the Minister has consulted the following sters in relation to the proposals—	39 40
		(a)	the Minister for the time being administering the <i>Public Works and Procurement Act 1912</i> ,	41 42
		(b)	the Minister for the time being administering the <i>Environmental Planning and Assessment Act 1979</i> .	43 44

30AD

	(3)	This section is repealed at the beginning of the day that is 12 months after the commencement of this section.	1 2			
30AE	Арр	lication of certain laws to development on designated land	3			
	(1)	The excluded laws do not apply to the following—	4			
		(a) the approval of the Minister to the carrying out of development approved by the Minister under section 30AD (the <i>approved development</i> ),	5 6 7			
		(b) the carrying out of the approved development by—  (i) Venues NSW, or	8 9			
		(ii) a person on behalf of, or with the authority of, Venues NSW,	10			
		(c) the use of the approved development from time to time,	11			
		(d) the designated land on which the approved development is carried out or used, or is proposed to be carried out or used.	12 13			
	(2)	In this section, excluded laws means—	14			
		(a) the Environmental Planning and Assessment Act 1979, and	15			
		(b) the Local Government Act 1993, and	16			
		(c) an instrument in force made under the Acts.	17			
30AF	Add	itional uses allowed on scheduled lands	18			
	(1)	Subject to section 30AI, the scheduled lands may be used for purposes permitted on the lands by a State environmental planning policy.				
	(2)	A State environmental planning policy may not permit the use of scheduled lands for a purpose unless the use of the land for the purpose has been approved by the Minister.	21 22 23			
	(3)	Section 30AE does not prevent a provision being included in a State environmental planning policy that applies to designated land.	24 25			
	(4)	If a provision included in a State environmental planning policy applies to designated land, section 30AE ceases to apply to the part of the land except in relation to development or use of the land in accordance with an approval under section 30AD—	26 27 28 29			
		(a) granted before the day on which the State environmental planning policy took effect, or	30 31			
		(b) granted on or after the day on which the State environmental planning policy took effect in response to an application made by Venues NSW before the day.	32 33 34			
	(5)	This section—	35			
		(a) does not affect a provision of a State environmental planning policy in force before the relevant day, and	36 37			
		(b) affects a provision included in a State environmental planning policy after the relevant day only to the extent the provision relates to the scheduled lands.	38 39 40			
	(6)	In this section—	41			
		relevant day means the commencement day of the Sydney Cricket and Sports Ground Amendment Act 2006.	42 43			
		State environmental planning policy has the same meaning as in the Environmental Planning and Assessment Act 1979.	44 45			

30AG	Variation or cessation of agreements relating to scheduled lands						
	(1)	To enable work to be carried out in relation to the scheduled lands, the Minister may determine an agreement relating to the scheduled lands—	3				
		(a) ceases to be binding on a party to the agreement, and	4				
		(b) is varied to the extent the Minister deems equitable.	5				
	(2)	The Minister must not make a determination on an agreement before receiving a report from Venues NSW that recommends the alteration of the agreement.	<del>7</del>				
	(3)	Notice of the Minister's determination must be given to the parties to the agreement in writing.	8				
	(4)	An agreement ceases to have effect or is varied—	10				
		(a) in accordance with the terms of the Minister's notice, and	11				
		(b) from the date specified in the Minister's notice.	12				
	(5)	In this section—	13				
		agreement includes a lease and a licence.	14				
30AH	Ancillary provisions relating to development and use of scheduled lands for additional purposes						
	(1)	The dedication of the scheduled lands for public recreation does not prevent or otherwise affect—	17 18				
		(a) the use of any part of the scheduled lands for a permissible purpose, or	19				
		(b) the grant of a lease or licence that permits or otherwise provides for its use for a permissible purpose.	20 21				
	(2)	To avoid doubt, a permissible purpose includes a purpose authorised by this Act for the purposes of section 2.12 of the <i>Crown Land Management Act 2016</i> in its application to the scheduled lands.	22 23 24				
	(3)	In this section, <i>permissible purpose</i> , in relation to a part of the scheduled lands, means a purpose permitted on that part by a State environmental planning policy referred to in section 30AF.	25 26 27				
30AI	Cert	ain uses of Venues NSW's land restricted	28				
	(1)	Despite any other provision of this Act, any other Act or any instrument made under this or any other Act—	29 30				
		(a) no part of Venues NSW's land, other than controlled land, may be used for residential accommodation, and	31 32				
		(b) no part of Venues NSW's land, other than designated land, may be used for tourist and visitor accommodation.	33 34				
	(2)	In this section—	35				
		<i>residential accommodation</i> means a building or place used predominantly as a place of residence, but does not include tourist and visitor accommodation.	36 37				
		tourist and visitor accommodation means a building or place that provides temporary or short-term accommodation on a commercial basis, and includes hotel accommodation and serviced apartments.	38 39 40				
30AJ	Dec	laration of controlled land or designated land	41				
		The Minister may, by order published on the NSW legislation website—	42				
		(a) declare certain land within Venues NSW's land to be—	43				

			controlled land, by inserting the description of the land into Schedule 4A, Part 3, or	1
			designated land, by inserting the description of the land into Schedule 4A, Part 2, or	3
	(b)		or abolish the classification of certain land within Venues land as controlled land or designated land by—	5
			amending the description of the controlled land in Schedule 4A, Part 3, or	7
		(ii) a	amending the description of the designated land in Schedule 4A, Part 2.	9
[15]	Part 5 Plans of m	anagem	nent	11
	Omit the Part.			12
[16]	Section 33A Adv	isory co	mmittees	13
	Omit the section.			14
[17]	Section 34 Use of	f Newca	stle Showground land	15
	Omit "the Author	ity" when	rever occurring. Insert instead "Venues NSW".	16
[18]	Section 34(5), de	finition	of "Authority"	17
	Omit the definition	n.		18
[19]	Section 37 Rang	ers		19
	Omit section 37(1	). Insert	instead—	20
			ecutive Officer may appoint a person employed by Venues NSW for the purposes of this Act.	21 22
[20]	Section 37(2)			23
	Insert ", and has t	ne power	rs and duties," after "functions".	24
[21]	Section 37(6) and	d (7) (def	finition of "Chief Executive")	25
	Omit "the Office	of Sport"	'wherever occurring. Insert instead "Venues NSW".	26
[22]	Section 40 Regu	lations		27
	Insert after section	n 40(1)(a)	.)—	28
	(a1)		otection of property, including buildings, structures and aery, situated on the authority's land, and	29 30
	(a2)		eservation or protection of trees or other vegetation on the ty's land, and	31 32
[23]	Section 40(1)(c1)	)		33
	Insert after section	n 40(1)(c	)—	34
	(c1)	areas of	nission to memberships that enable the access to, and use of, f Venues NSW's land reserved for the use of members, including ermination of membership fees or subscriptions, and	35 36 37
[24]	Section 40(1)(h1	)		38
	Insert after section	n 40(1)(h	u)—	39

			(h1)	the closing of lands, or a part of the lands, vested in or managed by a sporting venues authority, and the conditions that may apply to the closing of the lands or as a consequence of the closing of the lands, and	1 2 3
[25]	Sche	dules	1–3		4
	Omit	the So	hedul	es. Insert instead—	5
	Sch	nedu	le 1	Constitution and procedure of Board of Venues NSW	6
				section 15	8
	Par	t 1	Pre	liminary	9
	1	Inter	pretat	ion	10
			In th	is Schedule—	11
				d means the Board of Venues NSW.	12
				ber means a member of the Board and includes a non-GSE member.	13
				<b>GSE</b> member means a member of the Board appointed by the Minister r section 15(2)(a), (b) or (d).	14 15
	Par	t 2	Coi	nstitution	16
	2	Term	s of o	office of members	17
		(1)	Subje Boar	ect to this Schedule and the regulations, a non-GSE member of the d—	18 19
			(a)	holds office for the term, not exceeding 3 years, specified in the member's instrument of appointment, and	20 21
			(b)	is eligible, if otherwise qualified, for re-appointment.	22
		(2)		on-GSE member is not eligible to be a member of the Board for ecutive terms totalling more than 9 years.	23 24
	3	Part-	time a	appointments	25
			Non-	GSE members hold office as part-time members.	26
	4	Rem	unera	tion	27
			Servi	Minister may, from time to time, determine in consultation with the Public ice Commissioner an amount of remuneration, including travelling and istence allowances, in respect of the office of non-GSE members.	28 29 30
	5	Depu	ıties		31
		(1)	The 1	Minister may—	32
			(a)	from time to time, appoint a person to be the deputy of a non-GSE member, and	33 34
			(b)	revoke the appointment at any time.	35
		(2)		e absence of a non-GSE member, the member's deputy may, if available, a the place of the member.	36 37
		(3)		e acting in the place of the non-GSE member, a person has all the tions of the member and is taken to be a member.	38 39

	(4)	For the purposes of this clause, a vacancy in the office of a non-GSE member is taken to be an absence of the member.			
	(5)	This clause does not operate to confer on the deputy of a non-GSE member who is the Chairperson the member's functions as Chairperson.	3		
6	Vaca	ncy in office of member	5		
	(1)	The office of a member becomes vacant if the member—	6		
		(a) dies, or	7		
		(b) completes a term of office and is not re-appointed, or	8		
		(c) resigns the office by instrument in writing addressed to the Minister, or	9		
		(d) is nominated for election as a member of the Legislative Council or of the Legislative Assembly or as a member of a House of Parliament or a legislature of another State or Territory or of the Commonwealth, or	10 11 12		
		(e) is absent from 3 consecutive meetings of the Board of which reasonable notice has been given to the member personally or by post, except on leave granted by the Minister or unless the member is excused by the Minister for having been absent from the meetings, or	13 14 15 16		
		(f) becomes bankrupt, applies to take the benefit of any law for the relief of bankrupt or insolvent debtors, compounds with the member's creditors or makes an assignment of the member's remuneration for their benefit, or	17 18 19 20		
		(g) becomes a mentally incapacitated person, or	21		
		(h) is convicted in New South Wales of an offence that is punishable by imprisonment for 12 months or more or is convicted elsewhere than in New South Wales of an offence that, if committed in New South Wales, would be an offence so punishable, or	22 23 24 25		
		(i) is removed from office by the Minister under this clause.	26		
	(2)	The Minister may remove a member from office at any time.	27		
7	Chai	rperson and Deputy Chairperson	28		
	(1)	The Chairperson or Deputy Chairperson vacates office as Chairperson or Deputy Chairperson if they—	29 30		
		(a) are removed from the office by the Minister under this clause, or	31		
		(b) resign the office by instrument in writing addressed to the Minister, or	32		
		(c) cease to be a member of the Board.	33		
	(2)	The Minister may remove the Chairperson or Deputy Chairperson from office as Chairperson or Deputy Chairperson at any time.	34 35		
8	Fillir	g of vacancy in office of member	36		
		If the office of a member becomes vacant, a person is, subject to this Act and the regulations, to be appointed to fill the vacancy.	37 38		
9	Cası	ial vacancy in office of non-GSE member	39		
	(1)	A casual vacancy in the office of a non-GSE member is to be filled by a person appointed by the Minister.	40 41		
	(2)	The person who fills the casual vacancy holds office, subject to this Act, for the remainder of the term of the member who has vacated office	42		

	(3)	In this clause, <i>casual vacancy</i> means a vacancy that occurs other than because of the completion of a term of office.	1 2
10	Disc	osure of pecuniary interests	3
	(1)	A member must, as soon as possible after the relevant facts have come to the member's knowledge, disclose at a meeting of the Board—	4 5
		(a) that the member has a direct or indirect pecuniary interest in a matter being considered or about to be considered at the meeting, and	6 7
		(b) the nature of that interest.	8
	(2)	A member is not required to disclose an interest under subclause (1) if the interest does not appear to raise a conflict with the proper performance of the member's duties in relation to consideration of the matter.	9 10 11
	(3)	One or more of the following disclosures by a member at a meeting of the Board is a sufficient disclosure of the nature of the interest in any matter relating to a specified company or other body or to a specified person that may arise after the date of the disclosure and that is required to be disclosed under subclause (1)—	12 13 14 15 16
		(a) that the member is a member, or is in the employment, of that company or other body, or	17 18
		(b) that the member is a partner, or is in the employment, of that person, or	19
		(c) that the member has some other specified interest relating to that company or other body or that person.	20 21
	(4)	Particulars of any disclosure made under this clause must be recorded by the Board in a book kept for the purpose and that book must be open at all reasonable hours for inspection by any person on payment of the fee determined by the Board.	22 23 24 25
	(5)	After a member has disclosed the nature of an interest in any matter, the member must not, unless the Minister or the Board otherwise determines—	26 27
		(a) be present during a deliberation of the Board with respect to the matter, or	28 29
		(b) take part in any decision of the Board with respect to the matter.	30
	(6)	For the purposes of the making of a determination by the Board under subclause (5), a member who has a direct or indirect pecuniary interest in a matter to which the disclosure relates must not—	31 32 33
		(a) be present during any deliberation of the Board for the purpose of making the determination, or	34 35
		(b) take part in the making by the Board of the determination.	36
	(7)	A contravention of this clause does not invalidate any decision of the Board.	37
11	Effe	t of certain other Acts	38
	(1)	The provisions of the <i>Government Sector Employment Act 2013</i> relating to the employment of Public Service employees do not apply to a non-GSE member.	39 40
	(2)	A provision made by or under another Act requiring a person who is the holder of a specified office under that Act to devote the whole of his or her time to the duties of that office, or prohibiting the person from engaging in employment outside the duties of that office, does not operate to disqualify a person from—	41 42 43 44
		(a) holding the specified office and also the office of a member under this Act, or	45 46

		(b)	from accepting and retaining remuneration payable to the person under this Act as a member, if any.	1			
12	Pers	onal l	iability	3			
	(1)	A pi done	rotected person is not personally subject to any liability for anything	4 5			
		(a)	in good faith, and	6			
		(b)	for the purpose of executing functions under this Act.	7			
	(2)	The	liability instead attaches to the Crown.	8			
	(3)	In th	is clause—	9			
		done	e includes omitted to be done.	10			
			<i>lity</i> means civil liability and includes action, claim or demand.	11			
		•	ected person means—	12			
		(a)	the Board, or	13			
		(b)	a member of the Board, or	14			
		(c)	a person acting under the direction of the Board.	15			
Par	t 3	Pro	ocedure	16			
13	Gen	eral pr	rocedure	17			
			ect to this Act and the regulations, the Board may call and conduct tings of the Board as it determines.	18 19			
14	Quo	Quorum					
	(1)	The being	quorum for a Board meeting is a majority of its members for the time g.	21 22			
	(2)	is pr	bite subclause (1), the Board may, at a Board meeting at which a quorum esent, decide on a different number of members as the quorum for future tings.	23 24 25			
15	Pres	iding	member	26			
	(1)	The	Chairperson presides at a Board meeting.	27			
	(2)		e Chairperson is absent from a Board meeting, the Deputy Chairperson ides at the meeting.	28 29			
	(3)	meet	the Chairperson and the Deputy Chairperson are absent from a Board ting, another member elected by the members present at the meeting ides at the meeting.	30 31 32			
16	Voti	ng		33			
	(1)		ecision supported by a majority of the votes cast at a Board meeting at the a quorum is present is the decision of the Board.	34 35			
	(2)		ere is an equality of votes at a Board meeting, the member presiding at the ting also has a casting vote.	36 37			
17	First	meet	ing	38			
		The	Minister may call the first meeting of the Board in the manner that the	39 40			

18	Tran	saction of business outside meetings or by telephone	1
	(1)	The Board may, if it thinks fit, transact any of its business by the circulation of papers among all the members of the Board for the time being, and a resolution in writing approved in writing by a majority of the members is taken to be a decision of the Board made at a meeting of the Board.	2 3 4 5
	(2)	The Board may, if it thinks fit, transact any of its business at a meeting at which members, or some members, participate by telephone or other electronic means, but only if any member who speaks on a matter before the meeting can be heard by the other members.	6 7 8 9
	(3)	The Chairperson and each other member have the same voting rights as they have at an ordinary meeting of the Board for the purposes of—	10 11
		(a) the approval of a resolution under subclause (1), or	12
		(b) a meeting held in accordance with subclause (2).	13
	(4)	A resolution approved under subclause (1) is, subject to the regulations, to be recorded in the minutes of the meetings of the Board.	14 15
	(5)	Papers may be circulated among the members for the purposes of subclause (1) by electronic means.	16 17
Scł	nedu	Ile 2 Provisions relating to Chief Executive Officer of Venues NSW	18 19
		section 16	20
1	Арр	ointment of Chief Executive Officer of Venues NSW	21
	(1)	The Chief Executive Officer of Venues NSW—	22
		(a) holds office for the term, not exceeding 5 years, specified in the instrument of appointment, and	23 24
		(b) is eligible, if otherwise qualified, for re-appointment.	25
	(2)	A person is not eligible to be a Chief Executive Officer for consecutive terms totalling more than 10 years.	26 27
	(3)	The office of Chief Executive Officer is a full-time office and the holder of the office is required to hold it on that basis.	28 29
2	Rem	nuneration and allowances	30
	(1)	The employment of the Chief Executive Officer is, subject to this Act and the regulations, to be governed by a contract of employment between the Chief Executive Officer and the Minister.	31 32 33
	(2)	The office of Chief Executive Officer is a statutory office and the provisions of the <i>Government Sector Employment Act 2013</i> relating to the employment of Public Service employees do not apply to that office.	34 35 36
3	Vaca	ancy in office	37
	(1)	The office of Chief Executive Officer becomes vacant if the holder—	38
	, ,	(a) dies, or	39
		(b) completes a term of office and is not re-appointed, or	40
		(c) resigns the office by instrument in writing addressed to the Minister, or	41

		(d)	is nominated for election as a member of the Legislative Council or of the Legislative Assembly or as a member of a House of Parliament or a legislature of another State or Territory or of the Commonwealth, or	1 2 3
		(e)	becomes bankrupt, applies to take the benefit of any law for the relief of bankrupt or insolvent debtors, compounds with the Chief Executive Officer's creditors or makes an assignment of the Chief Executive Officer's remuneration for their benefit, or	4 5 6 7
		(f)	becomes a mentally incapacitated person, or	8
		(g)	is convicted in New South Wales of an offence that is punishable by imprisonment for 12 months or more or is convicted elsewhere than in New South Wales of an offence that, if committed in New South Wales, would be an offence so punishable, or	9 10 11 12
		(h)	is removed from office under clause 4.	13
	(2)		office of Chief Executive Officer becomes vacant, a person is, subject to act and the regulations, to be appointed to fill the vacancy.	14 15
4	Rem	oval fr	om office	16
			Minister may, in consultation with the Board of Venues NSW, remove the Executive Officer from office at any time.	17 18
5	Acti	ng Chie	ef Executive Officer	19
	(1)		Minister may, from time to time, and in consultation with the Board of es NSW, appoint a person to act in the office of Chief Executive Officer g—	20 21 22
		(a)	the illness or absence of the Chief Executive Officer, or	23
		(b)	a vacancy in the office of Chief Executive Officer.	24
	(2)		Minister may remove a person from office as acting Chief Executive er at any time.	25 26
	(3)	includ	cting Chief Executive Officer is entitled to be paid the remuneration, ding travelling and subsistence allowances, as the Minister may from to time determine.	27 28 29
Scł	nedu	le 3	Election of members of membership advisory committee by SCG Members	30 31
			section 21D(3)(b)	32
Par	t 1	Pre	liminary	33
1	Defi	nitions		34
		In thi	s Schedule—	35
			of nominations date means the date and time fixed by the Returning er for the close of nominations for an election.	36 37
		to be	on means an election, for the purposes of section 21D(3)(b), of persons e appointed as members of the membership advisory committee ituted under section 21D(1)(a).	38 39 40
			<b>rning Officer</b> means the person appointed by Venues NSW to be the ning officer for an election.	41 42

		<b>voting SCG Member</b> means a Member of the Sydney Cricket Ground who is of a class prescribed by the regulations.	1
2	Con	duct of election	3
		An election is to be carried out in accordance with this Schedule.	4
3	Retu	rning Officer	5
	(1)	Venues NSW is, by written instrument, to appoint a person as Returning Officer for the purposes of this Schedule.	6
	(2)	The Electoral Commissioner for New South Wales (or the Commissioner's delegate) may, if the Commissioner or delegate consents, be appointed as Returning Officer.	8 9 10
4	Voti	ng entitlements	11
	(1)	A voting SCG Member is entitled to vote in an election.	12
	(2)	Voting in an election is voluntary.	13
	(3)	If a voting SCG Member is an organisation, the voting rights of the Member may be exercised by a person authorised by the Member to vote on the Member's behalf.	14 15 16
	(4)	If a voting SCG Member is an organisation and holds more than one membership of a class prescribed in the regulations—	17 18
		(a) the voting SCG Member is entitled to one vote for each membership of the prescribed class held, and	19 20
		(b) the voting rights attached to each vote are to be exercised by a different person authorised by the voting SCG Member to vote on the voting SCG Member's behalf.	21 22 23
Par	t 2	Notice of elections and nomination of candidates	24
5	Peri	od for conducting election	25
	(1)	As soon as practicable after it has been determined in accordance with this Act that an election is required, Venues NSW is to notify the Returning Officer of the determination.	26 27 28
	(2)	An election is to be held no more than 90 days after the notification is given.	29
6	Noti	ce of election	30
	(1)	The Returning Officer is to cause a notice of an election—	31
		(a) to be published in the following—	32
		(i) the Gazette,	33
		(ii) at least one newspaper circulating generally throughout New South Wales, whether published in print or on a publicly accessible website,	34 35 36
		(iii) a prominent place on Venues NSW's website, and	37
		(b) to be sent by email to all voting SCG Members who have consented to receive notices and other documents relating to their SCG membership by email.	38 39 40
	(2)	The notice is to—	41
		(a) state that the election is to be held, and	42

		(b)	fix the polling day for the election, and	1
		(c)	invite nominations from voting SCG Members of candidates to fill the vacancies for elected members of the membership advisory committee, and	3
		(d)	fix the close of nominations date, being a date—	5
			(i) at least 21 days after the giving of the notice, and	6
			(ii) at least 30 days before the polling day for the election, and	7
		(e)	state the Returning Officer's contact details for the purposes of the lodgment of nominations.	8
	(3)	The elect	notice must be given at least 60 days before the polling day for the cion.	10 11
	(4)		Returning Officer may, by means of a further notice of the same kind, fix er close of nominations date than that fixed by a previous notice for the cion.	12 13 14
7	Nom	inatio	ns	15
	(1)	A no	omination of a candidate is to contain the following information—	16
		(a)	the full name of the candidate,	17
		(b)	the date of birth of the candidate,	18
		(c)	the residential address of the candidate,	19
		(d)	the membership card number of the candidate,	20
		(e)	a statement, signed by the candidate, in which the candidate consents to—	21 22
			(i) the nomination, and	23
			(ii) the use of the information contained in the nomination for the purposes of compiling a candidate information sheet,	24 25
		(f)	the full names, residential addresses, membership card numbers and signatures of at least 6 nominators, being voting SCG Members other than the candidate.	26 27 28
	(2)	emai	omination is to be in writing and may, if the Returning Officer specifies an il address for lodgment of nominations, be in the form of an electronic ment sent to the email address.	29 30 31
	(3)		omination is to be lodged with the Returning Officer before the close of inations date.	32 33
	(4)	A vo	oting SCG Member may nominate no more than 2 candidates in an ion.	34 35
8	No d	isplay	of material supporting candidates	36
		mate	erson must not post up, exhibit or distribute on the scheduled lands erial in support of a candidate's candidature at an election. imum penalty—1 penalty unit.	37 38 39
9	Unco	ontest	red elections	40
		If, af does	fter the close of nominations date, the number of candidates nominated not exceed the number of persons required for election, the Returning cer is to declare the nominated candidates elected.	41 42 43

Part 3		Contested elections	1
10	Whe	n poll to be taken	2
		If, after the close of nominations date, the number of candidates nominated exceeds the number of persons required for election, a poll is to be taken.	3 4
11	Prep	paration of roll	5
	(1)	If a poll is to be taken, the Chief Executive Officer of Venues NSW, or the Chief Executive Officer's delegate, is to certify a copy of the roll of all voting SCG Members and provide that copy to the Returning Officer.	6 7 8
	(2)	The roll is to contain the following information about each voting SCG Member—	9 10
		(a) the full name of the Member,	11
		(b) the residential address of the Member or other postal address specified by the Member for the purpose of receiving notices and other documents relating to the Member's SCG membership,	12 13 14
		(c) if the Member has consented to receiving notices and other documents relating to the Member's SCG membership by email—the email address specified for that purpose.	15 16 17
	(3)	The Returning Officer is to ensure that a copy of the roll is available for inspection free of charge by voting SCG Members.	18 19
	(4)	A voting SCG Member is to notify the Returning Officer of administrative errors relating to the Member's entry on the roll at least 14 days before the polling day for the election.	20 21 22
12	Prep	paration of candidate information sheet	23
	(1)	If a poll is to be taken, the Returning Officer is to compile a candidate information sheet about each nominated candidate.	24 25
	(2)	The candidate information sheet is to be based on—	26
		(a) information contained in the candidate's nomination, and	27
		(b) further information voluntarily lodged by the candidate with the Returning Officer before the close of nominations date and intended for inclusion in the candidate information sheet.	28 29 30
	(3)	Further information referred to in subclause (2)(b) is to be lodged as a statutory declaration in the form of the Eighth Schedule to the <i>Oaths Act 1900</i> , and may include the following—	31 32 33
		(a) the name of the candidate's employer or, if self-employed, the candidate's present position,	34 35
		(b) trade, academic or professional qualifications held by the candidate,	36
		(c) organisations of which the candidate is a member,	37
		(d) offices, other than employment, held by the candidate,	38
		(e) additional information relevant to the candidate's candidature, being no more than 4 typewritten lines.	39 40
	(4)	The Returning Officer may omit or rectify information, or reduce the length of information, from the candidate information sheet that the Returning Officer considers—	41 42 43
		(a) is not appropriate for inclusion in a candidate information sheet, or	44

		(b)	is misleading in any material particular, or	1
		(c)	is of a length greater than the length specified in subclause (3)(e).	2
	(5)		names of the candidates must be listed on the candidate information sheet e same order as they are listed on the ballot paper for the election.	3 4
13	Prep	aratio	n of ballot paper	5
		The	ballot paper must contain—	6
		(a)	the full names of all candidates nominated for the election, arranged in an order determined by the Returning Officer by drawing lots, and	7 8
		(b)	a box opposite and to the left of the name of each candidate.	9
14	Voti	ng pac	ck	10
	(1)	Ifaj	poll is to be taken, the Returning Officer must provide a voting pack to voting SCG Member at least 20 days before the polling day for the	11 12 13
	(2)	The	voting pack is to be provided—	14
		(a)	to a voting SCG Member who has consented to receive notices and other documents relating to the Member's SCG membership by email—by email to the email address specified for that purpose, and	15 16 17
		(b)	to any other SCG Member—by post to the Member's residential address or other postal address specified by the Member for the purpose of receiving notices and other documents relating to the Member's SCG membership.	18 19 20 21
	(3)	A vo	ting pack provided by post is to include the following—	22
		(a)	a ballot paper,	23
		(b)	a candidate information sheet,	24
		(c)	instructions for completing and lodging a ballot paper,	25
		(d)	one prepaid envelope addressed to the Returning Officer that contains, on its rear flap, spaces for the insertion of a voting SCG Member's name, address, signature and membership card number.	26 27 28
	(4)	A vo	ting pack provided electronically is to include the following—	29
		(a)	a ballot paper,	30
		(b)	a candidate information sheet,	31
		(c)	instructions for completing and lodging a ballot paper.	32
	(5)	Men	Returning Officer may, on written application made by a voting SCG aber, provide a duplicate voting pack to that Member if the Returning cer is satisfied that the original voting pack has been lost or destroyed.	33 34 35
15	Voti	ng pac	ck instructions	36
	(1)	comp	oting SCG Member who has received a voting pack may vote by pleting and lodging a ballot paper in accordance with the instructions ided in the voting pack.	37 38 39
	(2)		instructions for completing and lodging a ballot paper are to include the wing information—	40 41
		(a)	the polling day for the election,	42
		(b)	instructions on how to cast a valid vote by prepaid envelope,	43

		(c)	instructions on how to cast a valid vote electronically,	1	
		(d)	information about the preferential system applicable to the election,	2	
		(e)	any other information relating to the conduct of the election that the Returning Officer considers appropriate to include.	3 4	
	(3)		instructions on how to cast a valid vote by prepaid envelope are to include following further information—	5 6	
		(a)	instructions on how a voting SCG Member whose voting pack was provided electronically may obtain a prepaid envelope,	7 8	
		(b)	a statement that a ballot paper may be lodged, as an alternative to posting the ballot paper, by depositing the ballot paper in a sealed ballot box provided for that purpose at the office of Venues NSW.	9 10 11	
	(4)		instructions on how to cast a valid vote electronically are to include the wing further information—	12 13	
		(a)	the internet address of the voting website,	14	
		(b)	the information, including any applicable password, that a voting SCG Member will require to access the voting website.	15 16	
16	Electronic voting procedures				
		if th	voting website must provide a warning message to a voting SCG Member ere are errors on a ballot paper that the Member attempts to lodge ronically.	18 19 20	
17	Deat	h of c	andidate	21	
	(1)	day,	candidate dies after the close of nominations date and before the polling the election is taken to have wholly failed and the Returning Officer is to eed to give notice of a fresh election in accordance with clause 6.	22 23 24	
	(2)	advis	andidate dies on or after the polling day, but before the Returning Officer ses the Chief Executive Officer and the Minister of the result of the ion under clause 24, and the candidate—	25 26 27	
		(a)	is not elected—the candidate's preferences are to be distributed in accordance with clause 22, or	28 29	
		(b)	is elected—the candidate who obtained the next highest number of votes is to be declared elected in place of the deceased candidate, or	30 31	
		(c)	is elected with the highest number of votes and the election was for the purpose of electing 2 candidates—the candidate who scored the third highest number of votes is to be declared elected in place of the candidate who scored the second highest number of votes.	32 33 34 35	
Par	t 4	Ele	ction results	36	
18	Scru	tineer	s	37	
	(1)	A ca	ndidate may appoint, in writing, a scrutineer to represent that candidate.	38	
	(2)		tineers may be present during the inspection of prepaid envelopes, the all scrutiny of ballot papers and the counting of votes.	39 40	

(3)

A candidate may not act as a scrutineer.

19	Inspection of prepaid envelopes			1
	(1) The Returning Officer is to ensure that all prepaid envelopes purporting to contain ballot papers that are received before the polling day for an election are stored securely until they are inspected.			2 3 4
	(2)	On the polling day, the Returning Officer is to inspect each prepaid envelope and reject the envelope if—		5 6
		(a)	the envelope does not have legibly marked on its rear flap the name, address, signature and membership card number that appear to the Returning Officer to be of a voting SCG Member, or	7 8 9
		(b)	the envelope is not sealed.	10
20	Initia	l scrut	tiny of ballot papers	11
	(1)	On the polling day, the Returning Officer is to—		
		(a)	examine the contents of all prepaid envelopes that were not rejected under clause 19 and all ballot papers that were lodged by deposit in a sealed ballot box in accordance with clause 15(3), and	13 14 15
		(b)	review all ballot papers that were lodged electronically, and	16
		(c)	reject as informal any ballot papers that, in the Returning Officer's opinion, do not comply with the applicable instructions for casting a valid vote.	17 18 19
	(2)	Without limiting subclause (1)(c), the Returning Officer is to reject as informal any ballot paper—		
		(a)	in an election for the purpose of electing 2 candidates—if it does not have votes for 2 or more candidates indicated on it, or	22 23
		(b)	in an election for the purpose of electing 1 candidate—if it does not have a vote for 1 or more candidates indicated on it.	24 25
	(3)	If, in the Returning Officer's opinion, a person's voting intention is clearly indicated on the voting SCG Member's ballot paper, the Returning Officer is not to reject the ballot paper merely because the ballot paper contains an unnecessary mark.		26 27 28 29
	(4)		Returning Officer's determination of the formality or informality of a t paper is final.	30 31
21	Counting of votes			
	Following the initial scrutiny carried out under clause 20, the Returning Officer is to—			33 34
		(a)	count the votes recorded for each candidate on all ballot papers that were not rejected as informal in accordance with the preferential system applicable to the election, and	35 36 37
		(b)	count the informal ballot papers, and	38
		(c)	ascertain the results of the count, and	39
		(d)	inform the persons present of the result of the count.	40
22	Prefe	erentia	l system	41
	(1)	Ballo	t papers are to be counted by the Returning Officer—	42
		(a)	in an election for the purpose of electing 2 candidates—according to a multi-preferential system in which preferences 1 and 2 are regarded as	43 44

			primary votes and the remaining recorded preferences, if any, are regarded as secondary votes, or	2			
		(b	in an election for the purpose of electing 1 candidate—according to a preferential system in which preference 1 is regarded as a primary vote and the remaining recorded preferences, if any, are regarded as secondary votes.	3 2 5			
		(2) A	ll preferences recorded on ballot papers are fully transferable.	7			
			determination by the Returning Officer relating to the manner in which an ection is to be conducted is final.	9			
	23	Election	not invalid despite certain omissions	10			
		A	n election is not invalid merely because—	11			
		(a	a voting SCG Member whose name was on the roll prepared for the election did not receive a voting pack, or	12 13			
		(b	the Returning Officer did not receive a ballot paper lodged by prepaid envelope and sent to the Returning Officer before the polling day for the election.	14 15 16			
	24	Advice	of election result	17			
			ne Returning Officer is to advise the Chief Executive Officer and the inister of the result of the election within 7 days after the polling day.	18 19			
[26]	Sche	edule 4A		20			
	Insert after Schedule 4—						
	Sch	nedule	4A Venues NSW's land	22			
			section 3(1)	23			
	Par	t1 S	section 3(1)  cheduled lands				
		t 1 S		23			
	<b>Divi</b> All the	i <b>sion 1</b> hat piece o	cheduled lands	23 24			
	Divi All the of Sy of the	ision 1 hat piece o	Cricket and Sports Ground or parcel of land containing 11.208 hectares situate at Paddington in the City	23 24 25 26 27			
	Divi All the of Sy of the Divi	hat piece or ydney Pari at Parish. ision 2 hat piece of of Sydney	Cricket and Sports Ground or parcel of land containing 11.208 hectares situate at Paddington in the City sh of Alexandria County of Cumberland and being Portions 1528 and 1530	23 24 25 26 27 28			
	Divi All the of Sysof the Divi All the City Paris	hat piece or ydney Pari at Parish. ision 2 hat piece of of Sydney	Cricket and Sports Ground or parcel of land containing 11.208 hectares situate at Paddington in the City sh of Alexandria County of Cumberland and being Portions 1528 and 1530  Driver Avenue (etc) land or parcel of land containing 5851 square metres situate at Paddington in the Parish of Alexandria County of Cumberland and being Portion 1529 of that	23 24 25 26 27 28 30 31 32			
	Divi All the of Sy of the Divi All the City Paris Divi All the Course 2057	hat piece of ydney Parish.  ision 2 hat piece of Sydney sh.  ision 3 hat piece of Cunty of Cunty of Cunty of Cunty and for	Cricket and Sports Ground or parcel of land containing 11.208 hectares situate at Paddington in the City sh of Alexandria County of Cumberland and being Portions 1528 and 1530  Driver Avenue (etc) land or parcel of land containing 5851 square metres situate at Paddington in the	23 24 25 26 27 28 29 30 31			
	Divi All the of Sy of the Divi All the City Paris Divi All the Course 2057	hat piece of ydney Parish.  ision 2 hat piece of Sydney Sh.  ision 3 hat piece of Sydney Sh.  ision 3 hat piece of Cunty	Cricket and Sports Ground or parcel of land containing 11.208 hectares situate at Paddington in the City sh of Alexandria County of Cumberland and being Portions 1528 and 1530  Driver Avenue (etc) land or parcel of land containing 5851 square metres situate at Paddington in the Parish of Alexandria County of Cumberland and being Portion 1529 of that  Army Engineers Depot, Moore Park or parcel of land situated at Moore Park in the Parish of Alexandria and oberland containing 3.685 hectares, and being Lot 1 in Deposited Plan Normerly being the whole of the land in Certificate of Title Volume 9079 Folio	25 24 25 26 27 28 30 31 32 33 34 35 36			

commencing on the eastern side of Driver Avenue at its intersection with the northern side of a right of way shown in Sydney County Council Plan of Survey dated 27 August 1980 File Number S6343; and bounded thence on the west, southwest and northwest by the eastern, northeastern and southeastern sides of Driver Avenue bearing successively 356 degrees 20 seconds 165.48 metres, 355 degrees 36 minutes 68.37 metres, 308 degrees 41 minutes 114.89 metres, 322 degrees 20 minutes 20.115 metres, 346 degrees 20 minutes 20.115 metres, 4 degrees 50 minutes 20.115 metres and 18 degrees 20 minutes 75.74 metres to the intersection of the southeastern side of Driver Avenue with the southwestern side of Moore Park Road; on the northeast by the southwestern side of Moore Park Road bearing successively 103 degrees 22 minutes 30 seconds 50.27 metres, 106 degrees 8 minutes 95.72 metres, 110 degrees 16 minutes 30 seconds 60.13 metres, 113 degrees 4 minutes 27.62 metres, 113 degrees 30 minutes 22.53 metres, 114 degrees 32 minutes 7.15 metres, 115 degrees 40 seconds 21.54 metres, 116 degrees 34 minutes 20 seconds 21.33 metres, 117 degrees 44 minutes 30 seconds 18.18 metres, 119 degrees 56 minutes 20.45 metres, 121 degrees 56 minutes 40 seconds 32.105 metres, 123 degrees 41 minutes 18.3 metres, 125 degrees 29 minutes 30 seconds 22.755 metres, 126 degrees 51 minutes 30 seconds 15.405 metres, 125 degrees 23 minutes 13.72 metres and 122 degrees 24 minutes 19.065 metres to the northeastern corner of the land occupied, as at the date of assent to the Sydney Cricket and Sports Ground (Amendment) Act 1985, by the Department of the Army; on the southeast by a fenced line on the northwestern side of the bitumen sealed access walkway to the Sydney Showground bearing 217 degrees 37 minutes 40 seconds 144.4 metres to the corner of brick garages; along the southeastern face of those brick garages bearing successively 217 degrees 25 minutes 30 seconds 34.65 metres to the angle in the southeastern wall of the garages and a line bearing 231 degrees 46 minutes 30 seconds 39.91 metres; on the south by the southern face of a brick building and prolongation thereof bearing 262 degrees 27 minutes 14.23 metres and a line bearing 262 degrees 55 minutes 40 seconds 10.415 metres to the southeastern corner of the brick wall around the tennis courts, along the southern face of that brick wall to its southwestern corner bearing 262 degrees 12 minutes 30 seconds 37.77 metres and a line bearing 243 degrees 28 minutes 15.465 metres to the northern side of the Sydney County Council right of way aforesaid; generally on the southeast by lines along the generally northwestern sides of the right of way aforesaid bearing 265 degrees 52 minutes 5.8 metres, 262 degrees 29 minutes 7.16 metres, 259 degrees 21 minutes 6.875 metres, 256 degrees 28 minutes 6.955 metres, 253 degrees 1 minute 6.95 metres, 250 degrees 24 minutes 6.81 metres, 246 degrees 11 minutes 11.685 metres, 191 degrees 16 minutes 15.97 metres, 234 degrees 33 minutes 59.24 metres and 266 degrees 7.08 metres to the point of commencement and containing in all an area of 9.832 hectares or thereabout.

1

2

3

4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

29

30

31

32

33

34

35

36

37

38

39

40

41

42

43

44

45

46

47

48

49

50

51

52

53

#### Part 3 Controlled land

All that piece or parcel of land situated in the Parish of Alexandria County of Cumberland comprising part of Portion 1528 and part of Portion 1530 being part of the land in Schedule 2A: commencing on the southeastern side of Driver Avenue at its intersection with the southwestern side of Moore Park Road bounded thence on the northeast by the southwestern side of Moore Park Road bearing successively 102 degrees 6 minutes 50.27 metres, 104 degrees 51 minutes 30 seconds 95.72 metres, 109 degrees 60.13 metres and 111 degrees 47 minutes 30 seconds 23.475 metres to the intersection of the southwestern side of Moore Park Road with the northeastern face of a concrete kerb; on the northeast by the northeastern face of a concrete kerb and the prolongation thereof bearing 161 degrees 29 minutes 30 seconds 34.945 metres to the corner of a concrete pavement; on the southeast by a line joining the corner of a concrete pavement to the northeastern corner of a brick wall bearing 246 degrees 23 minutes 20 seconds 19.1 metres; along the southeastern face of a brick wall bearing successively 243 degrees 40 minutes 2.275 metres, 238 degrees 34 minutes 9.515 metres and 233 degrees 19 minutes 5.36 metres to the southeastern corner of a brick wall; by a line joining the southeastern corner of a brick wall to the northeastern corner of a brick wall bearing 230 degrees 52 minutes 10.745 metres; along the southeastern

face of a brick wall bearing successively 229 degrees 10 minutes 4.965 metres and 222 degrees 22 minutes 3.77 metres; by a line joining the southeastern corner of a brick wall to the northeastern corner of a brick wall bearing 222 degrees 11 minutes 2.305 metres; along the southeastern face of a brick wall bearing 222 degrees 24 minutes 14.915 metres; by a line joining the southeastern corner of a brick wall to the northeastern corner of a concrete kerb bearing 209 degrees 55 minutes 7.16 metres; along the southeastern face of a concrete kerb bearing successively 209 degrees 27 minutes 12.315 metres, 202 degrees 37 minutes 11.73 metres, 200 degrees 19 minutes 9.26 metres and 192 degrees 25 minutes 8.805 metres; along the southeastern face of a brick wall bearing 189 degrees 13 minutes 2.62 metres; on the south by the southern face of a brick wall bearing 277 degrees 45 minutes 4.53 metres; on the southeast by the southeastern face of a timber retaining wall bearing 252 degrees 25 minutes 4.03 metres; on the east by the eastern edge of a concrete dish drain bearing 186 degrees 36 minutes 14.26 metres; on the south by the southern face of a concrete kerb bearing 273 degrees 34 minutes 10 seconds 58.155 metres to the intersection of the kerb and the eastern side of Driver Avenue; on the southwest, west and northwest by the northeastern, eastern and southeastern sides of Driver Avenue bearing successively 307 degrees 24 minutes 30 seconds 114.89 metres, 321 degrees 3 minutes 30 seconds 20.115 metres, 345 degrees 3 minutes 30 seconds 20.115 metres, 3 degrees 33 minutes 30 seconds 20.115 metres and 17 degrees 3 minutes 30 seconds 75.74 metres to the point of commencement and containing in all an area of 3.2 hectares or thereabout.

1

2

3 4

5

6

7

8

9

10

11

12

13

14

15

16

17

18

19

20

21

22

23

24

25

26

27

28

#### Part 4 Venues

# Illawarra Region Lot 143 in Deposited Plan 786508 Lot 1 in Deposited Plan 1091530 Lot 6 in Deposited Plan 1091530 Lot 7 in Deposited Plan 1091530 Lot 96 in Deposited Plan 751299 Lot 101 in Deposited Plan 872951

### Newcastle Region

Newcastle Region	29
Lot 31 in Deposited Plan 1091407	30
Lot 32 in Deposited Plan 1091407	31
Lot 3238 in Deposited Plan 1013870	32
Lot 3239 in Deposited Plan 1013870	33
Lot 3241 in Deposited Plan 1013870	34
Lot 3242 in Deposited Plan 1013870	35
Lot 3243 in Deposited Plan 1013870	36
Lot 3244 in Deposited Plan 1013870	37
Lot 3245 in Deposited Plan 1013870	38
Lot 3235 in Deposited Plan 821124	39
Lot 7322 in Deposited Plan 1159253	40
Lot 3236 in Deposited Plan 821124	41
Lot 3284 in Deposited Plan 1049501	42
Lot 7302 in Deposited Plan 1143612	43
Lot 7303 in Deposited Plan 1143612	44
Lot 2498 in Deposited Plan 755247	45
Lot 2583 in Deposited Plan 755247	46

	Lot A in Deposited Plan 186497				
			a Region Deposited Plan 42643	2	
[27]	Schedule 5 Savings, transitional and other provisions				
	Insert after Part 5—				
	Part 6		Provisions consequent on enactment of Sporting Venues Authorities Amendment (Venues NSW) Act 2020	6 7 8	
	Divi	sion	1 Introduction	9	
	21 Inter		pretation	10	
		(1)	In this Part—	11	
			amending Act means the Sporting Venues Authorities Amendment (Venues NSW) Act 2020.	12 13	
			existing Venues NSW—see clause 24.	14	
			<i>instrument</i> means an instrument, except this Act or an instrument made under this Act, or any other document that creates, modifies or extinguishes rights or liabilities, or would do so if lodged, filed or registered in accordance with any law, and includes any judgment, order, process or other instrument issued by a court or tribunal.	15 16 17 18 19	
			<b>Local Venues Council</b> means an advisory committee that was, under clause 18(1) of this Schedule, taken to be established by the Minister under section 33A(2) as in force before the repeal day.	20 21 22	
			reconstituted Venues NSW—see clause 24.	23	
			<i>repeal day</i> means the day on which the <i>Sydney Cricket and Sports Ground Act</i> 1978 is repealed by this Act.	24 25	
			Sydney Cricket and Sports Ground Trust means the Sydney Cricket and Sports Ground Trust constituted by the Sydney Cricket and Sports Ground Act 1978 before its repeal.	26 27 28	
		(2)	If this Act provides for an event or other thing to occur on a particular day, that event or thing is taken to occur at the beginning of the day.	29 30	
	Division 2 Dissolution or reconstitution of certain bodies				
	22	Diss	olution of Sydney Cricket Ground Trust	32	
		(1)	On and from the repeal day—	33	
			(a) the Sydney Cricket and Sports Ground Trust is dissolved, and	34	
			(b) the Trust's assets, rights and liabilities immediately before its dissolution are transferred to reconstituted Venues NSW.	35 36	
		(2)	Division 4 applies to a transfer of assets, rights and liabilities under this clause.	37	
	23	Diss	olution of Local Venues Council	38	
		(1)	On and from the repeal day—	39	
		` '	(a) each Local Venues Council is dissolved, and	40	

		(b)	the Council's assets, rights and liabilities immediately before its dissolution are transferred to reconstituted Venues NSW.	1 2
	(2)	Divis	sion 4 applies to a transfer of assets, rights and liabilities under this clause.	3
24	Reco	nstitu	ition of Venues NSW	4
	(1)	On a	nd from repeal day—.	5
		(a)	Venues NSW as in existence immediately before the repeal day ( <i>existing Venues NSW</i> ) is taken to have been reconstituted by section 12 of this Act, as inserted by the amending Act, as a corporation with the same name ( <i>reconstituted Venues NSW</i> ), and	6 7 8 9
		(b)	the board of management of existing Venues NSW is dissolved.	10
	(2)	priva	Instituted Venues NSW is taken for all purposes, including the rules of the international law, to be a continuation of, and the same legal entity as, existing Venues NSW.	11 12 13
		Note.	As a result of this subclause, reconstituted Venues NSW retains all the assets, and liabilities of existing Venues NSW.	14 15
Divi	sion	3	Members of dissolved or reconstituted bodies	16
25	Abol	ition c	of certain offices	17
	(1)		of the following offices are abolished (an <i>abolished office</i> ) on and from epeal day—	18 19
		(a)	member of the Sydney Cricket Ground Trust,	20
		(b)	member of the board of management of existing Venues NSW,	21
		(c)	member of a Local Venues Council.	22
	(2)	Accordingly, a person who held an abolished office immediately before its abolition ceases to hold the office on and from the repeal day.		
	(3)	A person who ceases to hold an abolished office because of this clause—		
		(a)	is not entitled to any remuneration or compensation because of the loss of the office, and	26 27
		(b)	is eligible, if qualified, to be appointed as a member of the Board of reconstituted Venues NSW.	28 29
	(4)	Section 21D requires reconstituted Venues NSW to invite persons mentioned in subclause (1)(a) or (b) to become members of a membership advisory committee if they are not appointed members of the Board of reconstituted Venues NSW.		30 31 32 33
	(5)	A pe	rson invited to be a member of a membership advisory committee—	34
		(a)	is taken to be appointed for the balance of the term for which the person was appointed to the abolished office, and	35 36
		(b)	is entitled, despite section 21C(3), to payment of the remuneration and allowances, if any, determined by the Minister during the period.	37 38
Divi	sion	4	Transfers of assets, rights and liabilities	39
26	Application		n	40
			Division applies to the transfer of assets, rights or liabilities made by her provision of this Part if the provision provides for this Division to 7.	41 42 43

27	Inter	rpretat	iion	1
		are t	is Division, the person or body from which any assets, rights or liabilities ransferred is called the <i>transferor</i> and the person or body to which they ransferred is called the <i>transferee</i> .	2 3 4
28	Vest	ing in	transferee	5
	(1)		en any assets, rights or liabilities are transferred by a transfer to which this sion applies, the following provisions have effect—	6 7
		(a)	the assets of the transferor vest in the transferee by virtue of this clause and without the need for any further conveyance, transfer, assignment or assurance,	8 9 10
		(b)	the rights or liabilities of the transferor become by virtue of this Division the rights or liabilities of the transferee,	11 12
		(c)	all proceedings relating to the assets, rights or liabilities commenced before the transfer by or against the transferor or a predecessor of the transferor and pending immediately before the transfer are taken to be proceedings pending by or against the transferee,	13 14 15 16
		(d)	any act, matter or thing done or omitted to be done in relation to the assets, rights or liabilities before the transfer by, to or in respect of the transferor or a predecessor of the transferor is, to the extent to which that act, matter or thing has any force or effect, taken to have been done or omitted by, to or in respect of the transferee,	17 18 19 20 21
		(e)	the transferee has all the entitlements and obligations of the transferor in relation to the assets, rights and liabilities that the transferor would have had but for the transfer, whether or not the entitlements and obligations were actual or potential at the time the transfer took effect,	22 23 24 25
		(f)	a reference in any Act, in any instrument made under any Act or in any document of any kind to—  (i) the transferor, or  (ii) any predecessor of the transferor, to the extent to which the reference relates to the assets, rights or liabilities, is taken to be, or include, a reference to the transferee.	26 27 28 29 30 31
	(2)	The	operation of this Division is not to be regarded as—	32
	` ′	(a)	a breach of contract or confidence or otherwise as a civil wrong, or	33
		(b)	a breach of any contractual provision prohibiting, restricting or regulating the assignment or transfer of assets, rights or liabilities, or	34 35
		(c)	giving rise to any remedy by a party to an instrument, or as causing or permitting the termination of any instrument, because of a change in the beneficial or legal ownership of any asset, right or liability, or	36 37 38
		(d)	an event of default under any contract or other instrument.	39
	(3)	No a	attornment to the transferee by a lessee from the transferor is required.	40
29	No c	ompe	nsation payable	41
			compensation is payable to any person or body in connection with a sfer to which this Division applies.	42 43

43

30	Date	of vest	ing	1
			sfer of assets, rights or liabilities to which this Division applies takes on the day specified by the provision of this Part that gives rise to the er.	3
31	State	e taxes i	not chargeable	5
	(1)	State ta	ax is not payable in relation to—	6
		(a)	an exempt matter, or	7
			anything done because of, or for a purpose connected with or arising out of, an exempt matter.	9
	(2)	In this	clause—	10
		exemp	t matter means any of the following—	11
		]	the transfer of any assets, rights or liabilities by operation of this Division, including, without limitation, any instrument executed only for a purpose ancillary to or consequential on the operation of this Division,	12 13 14 15
			anything certified by the Minister in writing as having been done in consequence of the transfer, for example, the transfer or registration of an interest in land.	16 17 18
			ax means application or registration fees, duty or any other tax, fee or imposed by any legislation or other law of the State.	19 20
32	Con	firmatio	n of vesting	21
	(1)		linister may, by written notice, confirm a transfer of particular assets, or liabilities by operation of this Part.	22 23
	(2)	The no	otice is conclusive evidence of that transfer.	24
Divi	sion	5	Regulatory authorisations	25
33	Tran	sfer of o	certain regulatory authorisations	26
	(1)	Sydney	lause applies to a regulatory authorisation held by or on behalf of the y Cricket Ground Trust or a Local Venues Council (an <i>existing tory authorisation</i> ).	27 28 29
	(2)	reconst from the	isting regulatory authorisation becomes a regulatory authorisation of tituted Venues NSW (the <i>transferred regulatory authorisation</i> ) on and he repeal day for the purposes of the Act or statutory rule (the <i>relevant legislation</i> ) under which the existing regulatory authorisation was	30 31 32 33 34
	(3)	NSŴ	rovisions of the relevant State legislation apply to reconstituted Venues in relation to a transferred regulatory authorisation, subject to any cations prescribed by the regulations.	35 36 37
	(4)	of sub	inister must not recommend the making of a regulation for the purposes clause (3) unless the Minister administering the Act or statutory rule sed to be modified has consented to the modification concerned.	38 39 40
	(5)	State 1 transfe	on or body (a <i>regulatory body</i> ) that has the function under the relevant egislation of issuing regulatory authorisations of the same kind as a erred regulatory authorisation must, at the request of reconstituted s NSW, re-issue the transferred regulatory authorisation in the name of	41 42 43 44

			nstituted Venues NSW with substantially the same terms, conditions and presented as the transferred regulatory authorisation.	1 2
	(6)	for t	ee or charge is payable by reconstituted Venues NSW to a regulatory body he exercise of a function by the regulatory body in connection with the fer or re-issue of a regulatory authorisation by operation of this clause.	3 4 5
	(7)	In th	is clause—	6
	. ,	issue	ed includes given.	7
			latory authorisation means a licence, permit, consent, entitlement,	8
			editation or other authority under an Act or statutory rule, and includes, s not limited to, the following—	9 10
		(a)	a licence under the Liquor Act 2007,	11
		(b)	a licence under the Water Management Act 2000,	12
		(c)	a master licence under the Security Industry Act 1997,	13
		(d)	registration of a plant under the Work Health and Safety Regulation 2017,	14 15
		(e)	registration of a vehicle issued under the Road Transport Act 2013,	16
		(f)	a licence, permit, consent, entitlement, accreditation or other authority of a kind prescribed by the regulations.	17 18
Divi	sion	6	Employees	19
34	Emp	loyees	s of Sydney Cricket and Sports Ground Trust	20
	(1)	in th	persons employed under the <i>Sydney Cricket and Sports Ground Act 1978</i> e Sydney Cricket and Sports Ground Trust immediately before the repeal are transferred to the employment of reconstituted Venues NSW.	21 22 23
	(2)		unsfer of employment under this clause does not require the consent of the loyee transferred.	24 25
35	Emp	loyees	s of Public Service transferred to reconstituted Venues NSW	26
	(1)	perso to be NSW Note:	Minister may, by written order, transfer to reconstituted Venues NSW a on employed in a public sector agency who is designated by the Minister e a person required for the purposes of enabling reconstituted Venues V to exercise its functions.  Employees of existing Venues NSW continue as employees of reconstituted	27 28 29 30 31
	(2)		es NSW because of clause 24.	32
	(2)		he day specified in the order—	33
		(a)	the person ceases to be an employee of the public sector agency, and	34
		(b)	the person becomes an employee of reconstituted Venues NSW.	35
	(3)		unsfer of employment under this clause does not require the consent of the loyee transferred.	36 37
	(4)		following provisions apply in relation to the transfer of a person's loyment under this clause—	38 39
		(a)	the transfer has effect despite any other law, contract or instrument under a law,	40 41
		(b)	the transfer does not constitute a retrenchment, redundancy or termination of employment at the initiative of the Crown or public sector agency,	42 43 44

	(c)	the person transferred is not entitled to any payment or other benefit by reason only of having ceased to be an employee of a public sector agency as a result of the transfer,	1 2 3
	(d)	a public sector agency is not required to make any payment to the transferred person in relation to the transferred person's accrued rights in respect of annual leave, sick leave or extended or long service leave.	4 5 6
(5)		clause is repealed on the day that is 12 months after the commencement e clause.	7 8
(6)	In th	is clause, <i>public sector agency</i> means each of the following—	9
	(a)	the State, including the Crown in right of the State,	10
	(b)	a Minister,	11
	(c)	a State owned corporation within the meaning of the State Owned Corporations Act 1989,	12 13
	(d)	a public authority of the State,	14
	(e)	a wholly owned subsidiary of a public sector agency,	15
	(f)	any other person acting on behalf of the State or the Crown in right of the State.	16 17
Prov	/isions	s applying to transferred employees	18
(1)		following provisions apply in relation to an employee transferred to nstituted Venues NSW under this Division—	19 20
	(a)	the employee is entitled to continue as a contributor, member or employee for the purposes of any superannuation scheme in respect of which they were a contributor, member or employee, as an employee of the current employer, immediately before the transfer of employment and remains so entitled subject to any variation to that entitlement made either by agreement or otherwise in accordance with law,	21 22 23 24 25 26
	(b)	reconstituted Venues NSW is taken to be an employer for the purposes of any superannuation scheme in respect of which the employee continues as a contributor, member or employee pursuant to an entitlement under this clause,	27 28 29 30
	(c)	the employee retains their rights to annual leave, extended or long service leave or sick leave accrued or accruing immediately before the transfer of employment,	31 32 33
	(d)	the continuity of employment of the employee is not broken by the transfer of employment,	34 35
	(e)	service of the employee with the body from which the employee was transferred, including any service taken to be service as an employee, that is continuous service up to the time of the transfer of employment is, for all purposes, service with reconstituted Venues NSW,	36 37 38 39
	(f)	the conditions of employment applying to the employee immediately before the transfer of employment, whether under a State industrial instrument or contract of employment, continue to apply to the employee until the time that provision is otherwise made in accordance with any other Act or law.	40 41 42 43 44
(2)	prov	an employee transferred under clause 35, subclause (1)(f) extends to the isions, in force as at the repeal day, of regulations made under the ernment Sector Employment Act 2013 and the government sector	45 46 47

36

			oyment rules that confer or impose conditions of employment or gement on Public Sector employees in relation to the following matters—	1
		(a)	probation,	3
		(b)	security and other clearances,	4
		(c)	health clearances or assessments,	5
		(d)	requirements not to undertake other paid work without permission and to report charges and convictions for serious offences,	6
		(e)	absences from duty,	8
		(f)	the payment of increments,	9
		(g)	fitness for duty,	10
		(h)	the payment of allowances for temporary assignments to higher roles.	11
	(3)		he purposes of subclause (2), a reference in a provision of the regulations e rules referred to in the subclause—	12 13
		(a)	to a Public Service employee is taken to include a reference to an employee transferred under clause 35, and	14 15
		(b)	to the head of a Public Service agency or to the employer of a Public Service employee is taken to include a reference to reconstituted Venues NSW.	16 17 18
37	Oper	ation	of Commonwealth law	19
	(1)	conti	ovision of this Division, including a provision to the extent it imposes or nues a term or condition of employment, has no effect to the extent of any assistency with the <i>Fair Work Act 2009</i> of the Commonwealth.	20 21 22
	(2)	this NSW	void doubt, this clause does not limit the application of the provisions of Part in relation to reconstituted Venues NSW if reconstituted Venues V is declared to be a non-national scheme employer under section 9A of industrial Relations Act 1996.	23 24 25 26
Divi	sion	7	Miscellaneous	27
38	Refe	rence	to Schedule 3 in section 30A	28
		to be	Gerence in section 30A to Schedule 3 is taken, on and from the repeal day, a reference to Schedule 3 as in force immediately before the substitution e Schedule by the amending Act.	29 30 31
39			d persons under Sydney Cricket Ground and Sydney Football y-law 2014	32 33
		for a Foot	rson who was authorised by the Sydney Cricket and Sports Ground Trust function referred to in Part 6 or 7 the <i>Sydney Cricket Ground and Sydney ball Stadium By-law 2014</i> immediately before its repeal is taken, on and the repeal day, to have been appointed as a ranger under section 37 of this	34 35 36 37 38
40	Refe	rence	s to dissolved, abolished or reconstituted bodies or offices	39
	(1)	of th	ss the regulations provide differently, a reference in any instrument to any e following is, on and from the repeal day, to be read as including a ence to reconstituted Venues NSW—	40 41 42
		(a)	the Sydney Cricket and Sports Ground Trust,	43

(2)

(c)	a Local venue	s Coun	C11.						
The re	egulations may	make 1	provision	for or	with res	pect to h	ow refere	nces in	

1

2

3

instruments to bodies or offices dissolved, abolished or reconstituted by this Part should be read.

Sche	edu	le 2		Imendment of Sydney Olympic Park Authority act 2001 No 57	1 2
	Sche	edule (	5		3
	Insei	t after	Schedi	ule 4—	4
	Scł	nedu	le 5	Transfer of assets, rights and liabilities of Authority	5 6
	1	Inter	pretati	ion	7
		(1)	In thi	is Schedule—	8
		. ,	publi	ic sector agency means the following—	9
			(a)	the State, including the Crown in right of the State,	10
			(b)	a Minister,	11
			(c)	a State owned corporation within the meaning of the <i>State Owned Corporations Act 1989</i> ,	12 13
			(d)	a public authority of the State,	14
			(e)	any other person acting on behalf of the State or the Crown in right of the State.	15 16
		(2)		m used in this Schedule that is defined in Schedule 8, Part 3 has the same ting in this Schedule.	17 18
	2	Tran	sfer of	f specified assets, rights and liabilities by order	19
		(1)		Governor may, by written order, transfer to a public sector agency any s, rights and liabilities of the Authority specified in the order.	20 21
		(2)	The o	order is to be published in the Gazette.	22
	3	Vest	ing in	public sector agency	23
		(1)		eets, rights or liabilities are transferred by an order to which this Schedule es, the following provisions have effect—	24 25
			(a)	the assets of the Authority vest in the public sector agency by virtue of this clause and without the need for any further conveyance, transfer, assignment or assurance,	26 27 28
			(b)	the rights or liabilities of the Authority become by virtue of this Schedule the rights or liabilities of the public sector agency,	29 30
			(c)	proceedings relating to the assets, rights or liabilities commenced before the transfer by or against the Authority and pending immediately before the transfer are taken to be proceedings pending by or against the public sector agency,	31 32 33 34
			(d)	an act, matter or thing done or omitted to be done in relation to the assets, rights or liabilities before the transfer by, to or in relation to the Authority, to the extent to which that act, matter or thing has any force or effect, is taken to have been done or omitted by, to or in relation to the public sector agency,	35 36 37 38 39
			(e)	the public sector agency has the entitlements and obligations of the Authority in relation to the assets, rights and liabilities that the Authority would have had but for the transfer, whether or not the	40 41 42

		entitlements and obligations were actual or potential at the time the transfer took effect,	1 2
	(f)	a reference in any Act, in any instrument made under any Act or in any document of any kind to the Authority is taken to be, or include, a reference to the public sector agency to the extent to which the reference relates to the assets, rights or liabilities.	3 4 5 6
(2)	The	operation of this Schedule is not to be regarded as—	7
	(a)	a breach of contract or confidence or otherwise as a civil wrong, or	8
	(b)	a breach of a contractual provision prohibiting, restricting or regulating the assignment or transfer of assets, rights or liabilities, or	9 10
	(c)	giving rise to any remedy by a party to an instrument, or as causing or permitting the termination of any instrument, because of a change in the beneficial or legal ownership of an asset, right or liability, or	11 12 13
	(d)	an event of default under any contract or other instrument.	14
(3)	No a	ttornment to the transferee by a lessee from the transferor is required.	15
No c	ompe	nsation payable	16
	by ar	ompensation is payable to a person or body in connection with a transfer order to which this Schedule applies except to the extent, if any, to which rder giving rise to the transfer provides.	17 18 19
Tran	sfer o	f interests in land	20
(1)		order may transfer an interest in respect of land vested in the Authority out transferring the whole of the interests of the Authority in that land.	21 22
(2)	creat	e interest transferred is not a separate interest, the instrument operates to e the interest transferred under the terms that are specified in the ument.	23 24 25
(3)	This	clause does not limit any other provision of this Act.	26
Date	of ve	sting	27
		insfer of assets, rights or liabilities to which this Schedule applies takes t on the day specified in the order.	28 29
Cons	sidera	tion for vesting	30
	The	Governor may, by written order, specify—	31
	(a)	the consideration on which a transfer to which this Schedule applies is made, and	32 33
	(b)	the value or values at which the assets, rights or liabilities are transferred.	34 35
State	taxes	s not chargeable	36
(1)	State	tax is not payable in relation to—	37
	(a)	an exempt matter, or	38
	(b)	anything done because of, or for a purpose connected with or arising out of, an exempt matter.	39 40
(2)	In th	is clause—	41
	exen	apt matter means any of the following—	42

		(a)	the transfer of assets, rights or liabilities by operation of this Schedule, including, without limitation, an instrument executed only for a purpose ancillary to or consequential on the operation of this Schedule,	1 2 3
		(b)	anything certified by the Minister in writing as having been done in consequence of the transfer.	4
			e tax means application or registration fees, or other duties, taxes, fees or ges imposed under the legislation or other law of the State.	6 7
9	Con	firmati	on of vesting	8
	(1)		Minister may, by written notice, confirm a transfer of particular assets, s or liabilities by operation of this Schedule.	9 10
	(2)	The 1	notice is conclusive evidence of that transfer.	11
10	Effe	ct of tr	ransfer	12
			void doubt, this Act (apart from this Schedule) ceases to apply to an asset, or liability transferred by operation of this Schedule.	13 14

Sch	nedule 3	Amendment of other legislation	1
3.1	Criminal Pro	ocedure Regulation 2017	2
	Schedule 3 NS fees	W Government agencies and statutory bodies required to pay court	3 4
	Omit paragraph	(g) from the matter relating to the Department of Industry.	5
3.2	First State S	superannuation Act 1992 No 100	6
	Schedule 1 Em	ployers	7
	Omit "Sydney O	Cricket and Sports Ground Trust". Insert instead "Venues NSW".	8
3.3	Government	t Advertising Regulation 2018	g
	Clause 8 Exem pre-election pe	ption from restrictions on Government advertising campaigns during eriod	10 11
	Omit clause 8(2	(n).	12
3.4	Local Gover	nment Act 1993 No 30	13
	Section 556 WI and sewerage	hat land is exempt from all rates, other than water supply special rates special rates?	14 15
	Omit section 55	66(1)(m). Insert instead—	16
	(m)	land that is vested in the Crown or Venues NSW and is described in Schedule 4A, Parts 1, 2 or 3 of the <i>Sporting Venues Authorities Act 2008</i> and is used or occupied for the purposes of or in accordance with that Act,	17 18 19 20
3.5	Protection o	f the Environment Operations (General) Regulation 2009	21
	Clause 90 Outo	door entertainment activities	22
	Omit clause 90(	(1)(e). Insert instead—	23
	(e)	Venue's NSW land within the meaning of the Sporting Venues Authorities Act 2008,	24 25
3.6	Public Finar	nce and Audit Act 1983 No 152	26
	Schedule 2 Sta	atutory bodies	27
	Omit "Sydney O	Cricket and Sports Ground Trust".	28
3.7	Sporting Ve	nues Authorities Regulation 2019	29
[1]	Clause 5 Remo	oval of persons from sporting venues	30
	Insert after clau		31
	(c1)	contravening the Act or this Regulation, or	32
[2]	Clause 5(4A)		33
	Insert after clau	se 5(4)—	34

	(4A)	Venues NSW or a ranger may call in the aid of a police officer for the removal of a person from land, or a facility on land, vested in or managed by a sporting venues authority.	1 2 3
[3]	Clauses 7 Omit the cl		4
F41			,
[4]		dvisory committees on 33A(7)". Insert instead "21C(5)(d)".	-
[ <b>6</b> ]			,
[5]	Clause 9(2	e end of clause 9—	3
	(2)	This clause does not apply to a member who is appointed to a membership advisory committee under section 21D(3)(a) of the Act.	10 11
[6]	Clause 10	Local Venues Councils	12
	Omit the cl	ause.	13
3.8	Sporting	Venues (Invasions) Regulation 2016	14
	Clause 4 D	esignated sporting venues	15
	Insert at the	e end of the clause, with appropriate numbering—	16
		Sydney Cricket Ground at Driver Avenue, Moore Park,	17
		Sydney Football Stadium at Driver Avenue, Moore Park.	18
3.9	State Au	thorities Non-contributory Superannuation Act 1987 No 212	19
	Schedule 1	Employers	20
	~	ney Cricket and Sports Ground Trust" from Part 1.	21
	Insert instead	ad "Venues NSW".	22
3.10	State Au	thorities Superannuation Act 1987 No 211	23
	Schedule 1	Employers	24
	Omit "Sydr	ney Cricket and Sports Ground Trust" from Part 1.	25
	Insert instead	ad "Venues NSW".	26
3.11		vironmental Planning Policy (Exempt and Complying	27
	•	nent Codes) 2008	28
		27 Specified development	29
		edule 2 to the Sydney Cricket and Sports Ground Act 1978".	30
		ad "Schedule 4A, Part 1 of the Sporting Venues Authorities Act 2008".	31
3.12	State En	vironmental Planning Policy (Infrastructure) 2007	32
		Exempt development	33
	Omit "Sche	edule 2 to the Sydney Cricket and Sports Ground Act 1978".	34

	Insert instead "Schedule 4A, Part 1 of the Sporting Venues Authorities Act 2008".	1
3.13	State Environmental Planning Policy (State and Regional Development) 2011	2
	Schedule 2 State significant development—identified sites	4
	Omit "Part 1 of Schedule 2 to the <i>Sydney Cricket and Sports Ground Act 1978</i> " from clause 7(b).	5 6
	Insert instead "Schedule 4A, Part 1, Division 1 of the Sporting Venues Authorities Act 2008".	7
3.14	Water Savings Order 2005	9
3.14 [1]	Water Savings Order 2005 Schedule 1 Designated water users—businesses	g 10
	· ·	
	Schedule 1 Designated water users—businesses	10
[1]	Schedule 1 Designated water users—businesses Omit the matter relating to Sydney Cricket and Sports Ground Trust.	10 11